Author's response to reviews

Title: Analysis of human immune responses in longitudinal settings: Tutorial in Biostatistics

Authors:
Rajiv Sarkar (rsarkar@cmcvellore.ac.in)
Sitara R Ajjampur (sitararao@cmcvellore.ac.in)
Honorine D Ward (hward@tuftsmedicalcenter.org)
Gagandeep Kang (gkang@cmcvellore.ac.in)
Elena N Naumova (elena.naumova@tufts.edu)

Version: 3 Date: 27 January 2011

Author's response to reviews:

This revised version contains the reference to the Supplemental Table 1S with requested data. The manuscript by Ajjampur et al. is now accepted and referred in text as "in press". We will notify the Editor when the status change as soon it will be available on-line.